Author’s response to reviews

Title: Usefulness of a stool to stabilize dental chairs for cardiopulmonary resuscitation (CPR)

Authors:
Norimasa Awata (smile_dc_osaka@yahoo.co.jp)
Takashi Hitosugi (hitosugi.takashi.724@m.kyushu-u.ac.jp)
Yoichiro Miki (miki.yoichiro.485@m.kyushu-u.ac.jp)
Masanori Tsukamoto (tsukamoto@dent.kyushu-u.ac.jp)
Yoshifumi Kawakubo (y.kawakubo.710@gmail.com)
Takeshi Yokoyama (yokoyama@dent.kyushu-u.ac.jp)

Version: 4 Date: 24 Jul 2019

Author’s response to reviews:

Dear Editor and Reviewers (BMC Emergency Medicine)

Thank you very much for checking our manuscript and offering valuable advice. We have addressed remaining points and revised the manuscript accordingly.

1) If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

   I am sorry for my mistakes. I defined abbreviations which were used in the text in the text at first use and list of abbreviations provided in the Declarations. - Page 12 line 7-11.

2) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   We finally checked our manuscript about a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours, and tables/figures also be clean versions. Figures remained uploaded as separate files.